Author's response to reviews

Title: Socioeconomic deprivation and environmental noise exposure in Montreal, Canada

Authors:

Laura M Dale (laura.dale@mail.mcgill.ca)
Sophie Goudreau (sgoudrea@santepub-mtl.qc.ca)
Stéphane Perron (sperron@santepub-mtl.qc.ca)
Marianne Hatzopoulou (marianne.hatzopoulou@mcgill.ca)
Audrey Smargiassi (audrey.smargiassi@umontreal.ca)

Version: 2  Date: 22 September 2014

Author's response to reviews:

Dear Editor,

Here are the changes we have made to the manuscript:

The affiliation, email address of each author has been added to the first page and the corresponding author details.

The abstract was placed on the second page.

The conclusion and the acknowledgement were added at the end of the main text.

Yours sincerely,

Audrey Smargiassi (corresponding author)